Author's response to reviews

Title: A cohort study to evaluate persistence of hepatitis B immunogenicity after administration of hexavalent vaccines.

Authors:

- Cristina Giambi (cristina.giambi@iss.it)
- Antonino Bella (antonino.bella@iss.it)
- Antonella Barale (baraleantonella@asl20.piemonte.it)
- Domenico Montu (domenico.montu@asl17.it)
- Maria Marchisio (marchisio@asl19.asti.it)
- Maurizio Oddone (oddone@asl19.asti.it)
- Salvatore Zito (salvatore.zito@asl17.it)
- Maria Rapicetta (rapicett@iss.it)
- Paola Chionne (chionne@iss.it)
- Elisabetta Madonna (elisabetta.madonna@iss.it)
- Marta L Ciofi degli Atti (marta.ciofi@iss.it)

Version: 4 Date: 21 May 2008

Author's response to reviews:

I removed the qualifications from the submissions system and I modified the figure. I do not understand the first (Author list) of your indications, because it seems that the order of the authors is the same in the text and in the submissions system.

I also modified the format of the tables in the text.